Author’s response to reviews

Title: Unexplained chronic liver disease in Ethiopia: a cross-sectional study

Authors:
Stian Orlien (stian@orlien.no)
Nejib Ismael (nejibyusuf@gmail.com)
Tekabe Ahmed (tekaba321@yahoo.com)
Nega Berhe (nega.berhe.belay@gmail.com)
Trine Lauritzen (trine.lauritzen@vestreviken.no)
Borghild Roald (b.b.h.roald@medisin.uio.no)
Robert Goldin (r.goldin@imperial.ac.uk)
Kathrine Stene-Johansen (kathrine.stene-johansen@fhi.no)
Anne Dyrhol-Riise (a.m.d.riise@medisin.uio.no)
Svein Gundersen (s.g.gundersen@gmail.com)
Marsha Morgan (marsha.morgan@ucl.ac.uk)
Asgeir Johannessen (johannessen.asgeir@gmail.com)

Version: 2 Date: 29 Jan 2018

Author’s response to reviews:

Editor comment #1:
Please remove the cover letter and the response to reviewers from the inventory, as they are no longer necessary at this stage.

Author response:
The cover letter and the point-by-point response letter to reviewers have been removed from the inventory.

Editor comment #2:
At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Author response:

The suggested formatting changes have been made and a clean version of the manuscript and relevant tables/figures have been uploaded.